Author's response to reviews

Title: An Overlooked Connection: Serotonergic Mediation of Estrogen-Related Physiology and Pathology

Authors:

Leszek A. Rybaczyk (rybaczyk.1@osu.edu)
Meredith J. Bashaw (meredith.bashaw@fandm.edu)
Dorothy R. Pathak (pathak@msu.edu)
Scott M. Moody (moody@ohio.edu)
Roger M. Gilders (gilders@ohio.edu)
Donald L. Holzschu (holzschu@ohio.edu)

Version: 6 Date: 19 December 2005

Author's response to reviews:

19 December 2005

Editor, BMC Women's Health
RE: MS # 1696392487431583

To the Editor:

Enclosed is a revised version of the above referenced manuscript: An Overlooked Connection: Serotonergic Mediation of Estrogen-Related Physiology and Pathology. We have made the following changes to our manuscript:

* The country, USA, was added to the sixth author Dr. Roger Gilders

We appreciate the prompt, thorough, and professional review of the manuscript and feel that we have modified the text in accordance with the reviewer suggestions. We hope the paper will now be acceptable for publication. Thank you very much for your consideration. We look forward to hearing from you!

Sincerely,

Leszek A. Rybaczyk